Author's response to reviews

Title: Community acquired mastitis due to Mycobacterium abscessus: a case report

Authors:

Maria Bruna Pasticci Dr (pasticci@unipg.it)
Luigi Maria Lapalorcia Dr (luly1978@hotmail.it)
Giacomo Antonini Dr (giacomoantonini@yahoo.it)
Antonella Mencacci Dr (mencacci@unipg.it)
Rosanna Mazzolla Dott (rmazzolla@hotmail.com)
Franco Baldelli Dr (rprof2f1@unipg.it)

Version: 6 Date: 19 February 2009

Author’s response to reviews:

Dear Editor,

We re-submitted the revised manuscript

Compulsory:

1) The statement has been changed as follow: "The low dose of steroids administered one month prior to M. abscessus mastitis may have produced some negative immuno-modulatory effects predisposing her to infection"

Minor details:

1) Conclusion section has been added

2) Legends were present and arrows have been added

With best regards,

Maria Bruna Pasticci